Author’s response to reviews

Title: Experiences of pregnancy in adolescence of Internally Displaced Women in Bogotá: an ethnographic approach

Authors:

Yazmin Cadena Camargo (yaz_cadena@yahoo.com)
Anja Krumeich (a.krumeich@maastrichtuniversity.nl)
Maria Duque-Páramo (mcduhe@javeriana.edu.co)
Klasien Horstman (k.horstman@maastrichtuniversity.nl)

Version: 1 Date: 07 Dec 2018

Author’s response to reviews:

Dear Dr. Williams,

Thank you so much for your comments. We really appreciate them.

We adjust the manuscript according to your suggestions. We modified the background, the discussion and the ethical part. Our participants were all older than 18 years at the moment of the field work, even if they had their first child during adolescence. They all signed informed consent. (We explained more that part as well).

Best regards,

Yazmin Cadena Camargo.